Author's response to reviews

Title: Rhaphidophora korthalsii modulates peripheral blood natural killer cell proliferation, cytokine secretion and cytotoxicity

Authors:

Swee Keong Yeap (skyeap2005@gmail.com)
Abdul Rahman Omar (aro@lbs.upm.edu.my)
Wan Yong Ho (wanyongho@gmail.com)
Boon Kee Beh (boonkee06@gmail.com)
Abdul Manaf Ali (manaf@unisza.edu.my)
Noorjahan Banu Alitheen (noorjahan@biotech.upm.edu.my)

Version: 6 Date: 22 May 2013

Author's response to reviews: see over
22 May 2013

Dr. Tom Rowles  
Executive Editor  
BMC Complementary and Alternative Medicine

Dear Dr,

I wish to submit the revised regular article entitled "Rhaphidophora korthalsii modulates peripheral blood natural killer cell proliferation, cytokine secretion and cytotoxicity".

We have included below the response to the editorial comments.

**Comment**: Your ethics statement appears to have reverted to the version that appeared in your original submission. Could we ask you to check this and change it back to the version which confirms that specific approval for your study was granted by your institutional ethics committee.  
**Response**: We have revised the ethics statement as highlighted in page 5.

**Comment**: Please remove your figure legends from your uploaded figures, and instead include these at the end of your manuscript file, in accordance with our formatting guidelines.  
**Response**: Figure legends were move to the last part of manuscript according to the formatting guidelines.

I hope that the revised manuscript will be useful and suitable for publication. Your kind consideration is very much appreciated.

Best regards  
Noorjahan Banu Alitheen (PhD, ANU, Australia)  
Associate Professor  
Dept. of Cell and Molecular Biology  
Faculty of Biotechnology and Biomolecular Sciences  
Universiti Putra Malaysia  
43400 Serdang, Selangor  
Malaysia.

Tel: +603- 8946 7471  
Fax: +603-8946 7510  
Email: noorjahan@biotech.upm.edu.my